Author's response to reviews

Title: The relationship between non-alcoholic fatty liver and skeletal muscle mass to visceral fat area ratio in women with type 2 diabetes

Authors:

Xiaoyou Su (sxy186028@163.com)
Jing Xu (xujing_6@126.com)
Chao Zheng (zhengchao_3@163.com)

Version: 3 Date: 17 Jun 2019

Author's response to reviews:

Dear Editors and Reviewers:

Thank you for your letter and for the reviewers’ comments concerning our manuscript entitled “The relationship between non-alcoholic fatty liver and skeletal muscle mass to visceral fat area ratio in women with type 2 diabetes” (ID: BEND-D-19-00076R1). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and have made correction which we hope meet with approval. Revised portion are marked in red in the paper. The main corrections in the paper and the responds to the reviewer’s comments are as flowing:

Responds to the reviewer’s comments:

Reviewer:

1. In line 150, please correct "Yonden" (index) to "Youden", which is the correct spelling as the reviewer points out?

Response: we have revised it
2. Please have your manuscript carefully proofread to correct a number of English language issues in the manuscript

Response: we have revised it

We tried our best to improve the manuscript and made some changes in the manuscript. These changes will not influence the content and framework of the paper. And here we did not list the changes but marked in red in revised paper.

We appreciate for Editors/Reviewers’ warm work earnestly, and hope that the correction will meet with approval.

Once again, thank you very much for your comments and suggestions.

Corresponding author:

Name: Chao Zheng